Author's response to reviews

Title: The functional status and well being of people with myalgic encephalomyelitis/chronic fatigue syndrome and their carers

Authors:

Luis C Nacul (luis.nacul@lshtm.ac.uk)
Eliana M Lacerda (eliana.lacerda@lshtm.ac.uk)
Peter Campion (p.d.campion@hull.ac.uk)
Derek Pheby (derekpheby@btinternet.com)
Maria de L Drachler (jcdc.leite@googlemail.com)
Jose C Leite (jcdc.leite@googlemail.com)
Fiona Poland (F.Poland@uea.ac.uk)
Amanda Howe (Amanda.Howe@uea.ac.uk)
Shagufta Fayyaz (Shagufta.Fayyaz@lshtm.ac.uk)
Mariam Molokhia (Mariam.Molokhia@kcl.ac.uk)

Version: 7 Date: 20 May 2011

Author's response to reviews:

Dear BMC Series Editorial Production Team,

We are pleased to enclose the final version of the manuscript, addressing the required changes in its format, as shown below.

Major revisions:
1. Authors' contributions: it is now indicated in the text that all authors had the opportunity to read the final manuscript, and that all of them have approved the final version.
2. Tables: table 8 is re-orientated back into a landscape format.

Minor revisions:
1. Figures: we have tried to crop the figures as indicated in you guides, to exclude all the blank spaces around them, however, we are not sure if the uploaded pdf new versions will be in accordance to your requirements. In the case the figures are still not fitting well, we would be very grateful if you could make the required changes, as it was agreed in our phone conversation this afternoon.

We believe that with this final editorial changes the paper is now ready to be published.

Yours sincerely,

Dr Luis Nacul (on behalf of the authors)